Author’s response to reviews

Title: Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study

Authors:

Nobuyuki Nishikawa (nishikaw@kuhp.kyoto-u.ac.jp)
Rie Yago (yago@kc.twmu.ac.jp)
Yuichiro Yamazaki (yuichiro@gmail.com)
Hiromitsu Negoro (hnegoro@kuhp.kyoto-u.ac.jp)
Mari Suzuki (msuzuki@kc.twmu.ac.jp)
Masaaki Imamura (m69ima@kuhp.kyoto-u.ac.jp)
Yoshinobu Toda (toda@tenriyorozu-u.ac.jp)
Kazunari Tanabe (tanabe@kc.twmu.ac.jp)
Osamu Ogawa (ogawao@kuhp.kyoto-u.ac.jp)
Akihiro Kanematsu (aqui@hyo-med.ac.jp)

Version: 2
Date: 10 September 2014

Author’s response to reviews: see over
Ms Hazel Joyce Delos Santos  
Journal Editorial Office  
*BMC Urology*  

September the 10th, 2014

**MS ID.: 5131817871403085**  
Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study

Dear Ms Hazel Joyce Delos Santos  
Journal Editorial Office

Please find attached a revised version of our revised manuscript entitled “Expression of parathyroid hormone/parathyroid hormone-related peptide receptor 1 in normal and diseased bladder detrusor muscles: A clinico-pathological study.”

We have addressed the comments by The BMC Urology Editorial Team, and have revised the format in this version, as shown in the following pages.

We hope that in its current form the manuscript is suitable for peer review in *BMC Urology*.

Thank you again for your help and interest in our work.

Yours sincerely,

Akihiro Kanematsu MD, PhD  
Associate Professor  
Department of Urology, Hyogo College of Medicine  
1-1 Mukogawa-cho, Nishinomiya, Hyogo,  
663-8501 Japan
(1) Line Numbering:
Please revise your manuscript to include line and page numbers. Authors are asked to ensure that line numbering is included in the main text file of their manuscript at the time of submission to facilitate peer-review. Once a manuscript has been accepted, line numbering should be removed from the manuscript before publication. For authors submitting their manuscript in Microsoft Word please do not insert page breaks in your manuscript to ensure page numbering is consistent between your text.

We added line and page numbers in all the pages, and deleted a page breaks between the abstract and the manuscript.

(2) Consent:
Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

We added the following phrase as follows at Line 107.
“...and written informed consents for participation in the study were obtained from participants or parents.”